Author's response to reviews

Title: Enhanced virtual microscopy for collaborative education

Authors:

Marc M Triola (marc.triola@nyumc.org)
William J Holloway (william.holloway@nyumc.org)

Version: 4 Date: 25 January 2011

Author's response to reviews:

The abstract was copied and pasted into the boxes exactly as it appears in the manuscript. All vertical table lines and the vertical parts of the table border were made invisible.